Author's response to reviews

Title: Institutionalizing evidence-based practice: An organizational case study using a model of strategic change

Authors:

Cheryl B Stetler (cheryl.stetler@comcast.net)
Judith A Ritchie (judith.ritchie@muhc.mcgill.ca)
Jo Rycroft-Malone (hss401@bangor.ac.uk)
Alyce A Schultz (alyceme@cox.net)
Martin P Charns (mcharns@bu.edu)

Version: 2 Date: 15 May 2009

Author's response to reviews:

Thank you for the opportunity to respond to reviewers’ feedback on our manuscript. In the attached table, we have replied separately to the comments of each reviewer and indicated the location of manuscript changes made in response.